Introduction to Algorithms: 6.006 Massachusetts Institute of Technology Date: Professors Ron Rivest and Srini Devadas Handout 5

6.006 Feedback Form

Please give us some (anonymous) feedback so we can improve this course for future students.

• How are you doing in the class so far?

• How are we doing in the class so far?

• How can we improve lectures? How can we improve recitations?

• The pace of the class is: too slow · · · just right · · · too fast

• Is there any material you’ve especially liked or disliked? (Please indicate whether you like or dislike it.)

• What point or idea from the last week is most unclear to you?

• What else could we improve? Any other comments?